

FINAL REPORT

A SHARED VISION:

Ensuring quality of life for adults and children with asthma and allergies in First Nations and Inuit communities in Canada

Submitted:

January 15, 2009

To:

Assembly of First Nations/National Indian Brotherhood

By:

Asthma Society of Canada 4950 Yonge Street, Suite 2306 Toronto, ON M2N 6K1 1-866-787-4050 info@asthma.ca

Table of Contents

Table of Contents	1
Acknowledgements	2
Executive Summary	
Introduction and Background	
Organization of the Report	
Section 1: Research Design and Methods	
Section 2: Environmental Scan	
Population Characteristics	
Population Distribution	
Living Situation	
Burden of Illness	
Summary	
Section 3: Results from Interview Data Analysis	
Community Health	
Community Resources	
Burden of Illness	
Community Expertise	
Section 4: Results of Focus Group Analysis	
Methods	
Key Themes	20
Theme 1- Lack of Access to Culturally Appropriate Asthma/Allergy Care and	
Education	20
Theme 2- Acknowledging the Risk	23
Theme 3- Lack of Community Empowerment	25
Section 5: Conclusions and Recommendations	27
References	
APPENDICES	35
Appendix 1: Survey Questions	35
Appendix 2: Focus Group Questions	37

Acknowledgements

This report was made possible through the generous contributions of a number of groups and individuals. We thank them for providing their input, sharing their experiences, expertise and feedback.

First, we would like to thank the members of the working group: Kim Barker, Sarah Carriere, Onalee Randell, Kathy Green, Susan Elliott, Joanne Gillespie, Heidi Liepold, Pauline LeFebvre, and Oxana Latycheva. Your input was invaluable and your guidance was integral to the completion of this project.

The initial project manager, Kim Bulger, and her team Anu Singh, Russell Sheridan, Dwayne Unru, Zoran Vidac, Ialeen Jones, and Matila Atleo were instrumental in undertaking data collection, analysis, and report writing.

In the later stages of the work, project manager, Susan Elliott and her research assistant, Michelle Vine, provided descriptive analysis and written contributions, as well as made presentations to key stakeholders.

We thank James Patterson for conducting an environmental scan and a literature review in the early stages of the project. From the Asthma Society of Canada specific thanks go to Meridene Haynes, Zhen Liu and Christine Hampson.

Malcolm King, Scientific Director of the Institute of Aboriginal Peoples' Health, Canadian Institutes of Health Research (CIHR) provided invaluable input and advice on an earlier draft of the report.

French translation of the report was provided by Olivier Depenweiller from Alpe Translations.

Finally, we would like to thank the individuals and groups who participated in this research. Specifically, we thank Dr. Mark Greenwald, Chair of the Medical and Scientific Committee of the Asthma Society of Canada, representatives of the Assembly of First Nations, Inuit Tapiriit Kanatami, the Public Health Agency (PHAC) and First Nations and Inuit Health Branch – Health Canada (FNIHB) and the AllerGen National Centre of Excellence for allergies and asthma.

Executive Summary

The prevalence of asthma and related allergies in Canada and much of the western industrialized world has increased substantially in the last few decades. In the development of a National Lung Health Framework for Canada, key stakeholders are concerned that the health needs of Canada's First Nations and Inuit peoples are addressed. These concerns relate specifically to the prevalence of asthma and related allergies in these populations. While the data are not as robust as we would like them to be, there are preliminary indications that the overall prevalence may be higher. Even if the prevalence of allergies among First Nations children is similar to that of children in the general population in Canada (12.2% and 16.4%, respectively), less than one third of First Nations children with allergies are receiving treatment. Issues of access to health care therefore may complicate the situation further as First Nations children and families try to cope with and seek treatment for asthma and related allergies. It is difficult to assess the extent of the circumstances because there is a lack of Inuit data in general, and specifically related to asthma and related-allergies. This exploratory study has examined the perceptions and experiences of asthma and related allergies in First Nations and Inuit communities in Canada.

Three sources of data were used to compile the findings in this report. First, an environmental scan was conducted of relevant public documents that would allow for a comprehensive description of the populations of interest. Second, focus groups were conducted in 3 First Nations and 3 Inuit communities to explore the impacts and experiences of asthma and related allergies, as well as gaps in existing services. Third, telephone interviews (n=30) were conducted with relevant community health personnel (2 per each First Nations and Inuit region). Important findings resulting from this report include prevalence, impacts and the need for management strategies tailored to these communities. These data have shown us that there are major gaps in family and community resources for coping with and managing asthma and related allergies. Further, there appears a lack of health care resources, opportunities and training; not surprising due to a range of access issues, including geographic access.

The findings from this analysis support five key recommendations, providing the ground work for further research in the area of First Nations and Inuit asthma and related allergies. In the context of the National Lung Health Framework, the development of culturally appropriate education and community asthma/allergy awareness is integral for improving access to asthma/allergy-related health services. Secondly, there is an acute need for a federal framework of asthma and allergy care for health care providers. In order to improve the level of diagnosis, care and management, physicians, nurses and community health workers should be encouraged to apply government asthma/allergy guidelines specific to Inuit and First Nations communities. Thirdly, key stakeholders must work together to develop a new health care delivery model for First Nation and Inuit people living with chronic conditions. A fourth recommendation involves addressing major gaps in family and community resources. Such a recommendation would encompass the implementation of programs for chronic disease prevention, health

promotion, injury prevention, infectious disease control, tobacco control and prevention, prenatal care, nutrition and obesity, and programs for asthma, lung health and related-allergies. Finally, it is essential that we begin to address the link between obesity and asthma and related-allergies by way of school-based programming.

While these recommendations are based on the evidence gathered to undertake this report, a clear take home message is the need for more research, better data, increased monitoring and surveillance, and enhanced access to care.

Introduction and Background

This report is based on the findings of a research project sponsored by the Asthma Society of Canada, in collaboration with the Assembly of First Nations and the Inuit Tapiriit Kanatami. It is intended to provide recommendations for a strategy to address the burden of illness that asthma and associated allergies represent for First Nations and Inuit communities in Canada, given that there is widespread concern that they experience respiratory illness at rates higher than the Canadian population. While it may seem straightforward to report prevalence rates for asthma, it is actually rather difficult due to issues around diagnosis, lack of monitoring and surveillance, etc. It is even more difficult to provide accurate prevalence data for Canada's First Nations and Inuit peoples' issues in some cases of isolation and access to care. As a result, reported prevalence rates vary. For example, according to the Canadian Community Health Survey, asthma prevalence for the Canadian population hovers around 8%, with a rate of 13% among children 11 years and younger. However, a composite measure reported by key researchers in the area puts the range closer to 15 - 20% (Subbarao et al, 2009). While the Aboriginal Peoples Survey (2001, which includes both on and off reserve First Nations, Inuit and Metis) reports asthma levels at around 12% overall, the First Nations Regional Health Survey, 2002-03 (for on reserve only) reports an overall rate of around 11%; 15% for youth and 14% for children under 12. Finally, Statistics Canada (2007) puts the rate among our Inuit population at approximately 6% overall and levels in children similar to the Canadian population as a whole however issues of access to health care in the Arctic may be impacting these prevalence rates. Indeed, approximately 28% of Inuit children (aged 0 - 14) have contact with a family doctor at least once per year compared to 67% of children in the general population in Canada. Comparing these data across groups becomes even more difficult when we examine the wording of the question asked to elicit the response. For example, while the Canadian Community Health Survey asks about physician diagnosed asthma, the First Nations Regional Health Survey asks if a respondent has ever been told they have asthma, with no attribution specified. In summary, rates vary but the take home message remains the same: rates are concerning and potentially on the rise; further, these populations may be more at risk than others due to their exposure to well known determinants of respiratory illness, as well as documented access to care issues.

In April 2006, the Canadian Lung Association initiated the development of a National Lung Health Framework. This is in the context of the burden of illness that respiratory disease represents to Canadians. Indeed, the World Health Organization estimates that by

the year 2020, lung disease will be the third leading cause of death in the world. For Canada to respond to this impending health crisis, a comprehensive and coordinated national plan was required that could attempt to manage the social, economic, cultural and health implications of this burden. The resulting framework represents a coordinated approach to prevention and management of respiratory disease and is intended to inform policy, research, innovation, education and leadership in the area of Canadian lung health.

The findings presented in this report will feed into the Lung Health Framework, as well as inform health care providers, asthma educators and the general public about issues of illness and access with respect to asthma and associated allergies among Canada's First Nations and Inuit peoples in order to actualize a radical improvement in the quality of life of those impacted.

Organization of the Report

The report is divided into five sections. Section one briefly describes the methods used to undertake the collection of data and information reported on in the document. Section two describes the results of an environmental scan of key documents that allow a comprehensive description of the populations of interest (i.e., demographics, geographic distribution, living situation, burden of illness). Section three links determinants with the populations of interest while section four presents the results of focus groups and interviews with key informants in both First Nations and Inuit communities about the experiences and burdens associated with asthma and allergies. Section five concludes the report with recommendations linking back to the National Lung Health Framework.

Section 1: Research Design and Methods

The Asthma Society of Canada worked with the project Steering Committee to identify a researcher that had both the network and the skills to undertake the study as described. Kim Bulger was recommended and thus chosen as the lead on the project, and mandated with the task of collecting the data.

Three sources of data were used to compile the findings in this report. First, an environmental scan was conducted of relevant public documents that would allow for a comprehensive description of the populations of interest. This information includes demographics, geography, living situation, and burden of illness. Second, 6 focus groups were undertaken in the spring of 2008 in 3 First Nations and 3 Inuit communities in order to explore the impacts and experiences of asthma and related allergies, as well as gaps in existing services. Focus group participants included members of advisory committees, health care providers, parents and school principles. Third, telephone interviews (n=30) were conducted with relevant community health personnel (2 per each First Nations and Inuit region). This sample list was provided by the Assembly of First Nations (AFN) and Inuit Tapiriit Kanatami (ITK) representatives. If someone had left the organization, the

organization suggested an alternative. All interviews were conducted in the spring of 2008 by a single researcher and lasted approximately 30 minutes. All questions were open-ended.

Descriptive analyses of the quantitative survey data and qualitative analyses of the qualitative data were undertaken by researchers at McMaster University. Oral presentations of preliminary results were made to both the AFN and ITK in order to assess their sense of whether or not the results resonated with their understanding of how these issues were perceived and experienced in their relative communities. On the whole, their view was that the results 'made sense', with a couple of caveats. There was a sense among the ITK that asthma and related allergies would have emerged as a health issue of higher priority if the question has been targeted to children and youth. Second, the AFN felt strongly that results had to reflect the importance of cultural practices; for example, the reliance on wood burning as a source of home heating.

The research has been overseen by a working group made up of staff from the Asthma Society of Canada, representatives of the Assembly of First Nations, Inuit Tapiriit Kanatami, the Public Health Agency of Canada (PHAC) and the First Nations and Inuit Health Branch – Health Canada (FNIHB) and the AllerGen National Centre of Excellence.

It is important to note that, while informative, the study sample is not large and therefore can not represent the views of all relevant stakeholders. Having said that, the breadth and representativeness of the sample group provides a solid foundation for thinking about issues of respiratory health among First Nations and Inuit communities in Canada, as well as a way forward.

Section 2: Environmental Scan

Population Characteristics

The growth rate among Aboriginal (First Nations, Inuit and Metis) Peoples in Canada (1.8% per annum) is much higher than that of the non-Aboriginal population (less than 1%). In 2001, the Aboriginal population (of approximately 1 million) represented 3.4% of Canada's total population. By 2017, projections indicate Aboriginal Peoples in Canada will represent over 4% of the national population. The biggest contributing factor to population growth is fertility, given that the current Aboriginal birth rate is 1.5 times the overall Canadian rate. The majority (approximately two-thirds) of Aboriginal Peoples in Canada are First Nations with Metis representing about 27% and Inuit about 5% (Statistics Canada 2005). According to a medium growth scenario, the Inuvialuit population will have the fastest rate of growth (approximately 2.3%).

In 2006, 2.2% of the Canadian population self-identified as First Nations (approximately 700,000 individuals) while approximately 50,000 identified as Inuit; this is an increase of

26% from 1996. Nunavik is the fastest growing Inuit region (25% increase since 1996) while Inuvialuit experienced a slight decline (3%) during the same time period.

The mean age of the First Nations population at the time of the last census was 25 years, compared to 22 years for the Inuit population and 40 years for the general Canadian population. Further, 31% of the First Nations and 56% of the Inuvialuit population were under the age of 24 years with only a very small percentage (about 4%) above age 65 (compared to 13% of the general Canadian population). Life expectancy for the First Nations population in Canada in 2006 was 69 years for men and 77 years for women; 63 and 72 years for the Inuvialuit population. This is low, compared to the general Canadian population (77 and 82 years for men and women, respectively). Among Inuit –inhabited populations life expectancy has decreased by one year (68 vs 67) during the period 1991 to 2001; this figure parallels the overall life expectancy rates in Canada in 1946 (Wilkins et al. 2008).

The proportion of First Nations people who can speak a native language well enough to carry on a conversation is relatively low (29%) but has remained steady from 1996 to 2006. Very few First Nations people did not speak English (1% among total population; 5% among seniors). In 2006, 64% of Inuit reported Inuktitut as their mother tongue, down 4% from 1996.

The combination of high fertility rates and increasing life expectancies means that the First Nations and Inuit populations will continue to have increasing numbers of young people, but these young people will have a larger cohort of 'dependents' to look after (infants and children; seniors). Prevalence of asthma and allergies will likely increase in these populations under this scenario, while related respiratory illness (e.g., Chronic Obstructive Pulmonary Disease) may increase among the older segment of the population, particularly smokers.

Population Distribution

83% of the First Nations population in Canada lives in Ontario, Manitoba, Saskatchewan, Alberta and BC. Less than half the First Nations population lives on reserve; 98% of these are Status Indians. The diffuse nature of the distribution of this population may affect access to appropriate health care, as well as educational and other services related to asthma and related allergies, in addition to other health outcomes.

78% of the Inuit population in Canada is spread across four regions within Inuit Nunaat which stretches from Labrador to the Northwest Territories: Territory of Nunavut (49%), Nunavik (19%), Inuvialuit region (6%) and Nunatsiavut (4%). The dispersed nature of this population in remote areas affected by severe weather for much of the calendar year has obvious implications for access to health care services.

Living Situation

Most First Nations children live with a relative, but are much more likely to live with a lone parent, grandparent or other relative (45%) than is the case in the general Canadian population (18%). 69% of Inuit children are living with both parents, compared with 54% of the First Nations population and 82% in the rest of Canada. A partial explanation may stem from the unique cultural practices in Inuit communities whereby informal adoption is common.

15% of the First Nations population resides in a crowded dwelling according to the 2006 census compared to 3% of the general Canadian population. A *crowded dwelling*, according to Statistics Canada, is defined as more than one person per room. Overcrowding is more common on reserve, where 26% of the population lives in overcrowded conditions (down from the 33% reported in the 1996 census). Off-reserve, 7% of First Nations live in overcrowded conditions and 31% of Inuit live in overcrowded dwellings according to the 2006 census. However, it is important to note that Inuit are four times more likely to share a dwelling with one or more families.

The census asks Canadians to report whether or not they feel their home is in need of major repairs. According to the 2006 census, the rate of homes in need of major repairs in the non-Aboriginal population was 7%. The comparable rate for First Nations was 44% (on reserve, up from 36% in 1996) and 17% (off reserve). 28% of Inuit peoples reported their home was in need of major repairs, an overall increase from 1996.

Additional data on living situation comes from the 2002/2003 Regional Health Survey undertaken among First Nations communities in Canada. Herein we learn that 22% of First Nations report no access to garbage collection, while another 9% report no access to septic or sewage services in the home. Only two-thirds of respondents to this survey felt their water was safe to drink.

All of these living situations have been shown to be determinants of health in some contexts and populations. Lone parenting can be stressful and that can impact health; so too can the conditions of overcrowding which can also affect the transmission of infectious diseases. Overcrowding can lead to the transmission of respiratory tract infections in children, resulting in the increased likelihood of developing chronic respiratory tract diseases (e.g., asthma). These situations may lead to increased allergic sensitization, increased incidence of asthma, and increased incidence of the exacerbation of asthma. Obviously, poorly maintained dwellings can lead to several health situations, both physiological and psychosocial. Older dwellings and those with dust, damp and mould have been shown to be associated with asthma and related allergies.

Burden of Illness

Asthma is a chronic inflammatory condition of the airways characterized by recurrent symptoms of variable airflow limitation. Allergy is an important factor in lung health. Allergic rhinitis (ie. hay fever) and asthma very often occur together. Atopic dermatitis is

often seen in young children who later develop asthma (Leung 2004). The risk factors for, or determinants of, asthma and related allergies (e.g., chronic rhinitis) have become the subject of much attention by researchers given substantial increases in disease prevalence over the past few decades. This research has shown firstly that there is substantial geographic variation in asthma prevalence; for example, rates in some parts of the world (e.g., Asian countries) are much lower (2-4%) than in other parts of the world where rates are in the range of 15 - 20 % (e.g., UK, Canada, Australia, New Zealand). Secondly, no clear risk pattern emerges for asthma causation, although a wide range of factors have been shown to be either a determinant of or a protective mechanism for asthma and allergies. Sometimes this literature can be very confusing because of conflicting reports. For example, for many years, researchers suggested that breastfeeding of infants had a protective effect for asthma and allergies. Recent studies have shown that in fact this may not be the case. The same is true for the presence of pets in the household; some studies show that pets can exacerbate lung diseases, while others indicate they may have a protective effect. Thirdly, there is extensive discussion in the literature about the relative contribution to the 'asthma epidemic' of genes versus environment. The current state of thinking indicates quite rightly that the relative recency of the asthma epidemic would rule out 'genes' as the primary contributing factor; in brief, human genes just don't change that fast. This conclusion leads us to point the finger at environment; indeed, it plays a major role, but given that genetics does play some role (i.e., asthma and allergies are much more prevalent in children of parents with asthma or allergy), researchers now suggest that genes and environment are working together to affect respiratory health, mediated by both innate and adaptive immunity in humans. This relationship is further complicated by the fact that environmental triggers may impact the development of asthma and related allergies at different times in life and the relevant risk factors may change over time (Subbarao et al., 2009).

In short, there is a lot we do not know about the major risk factors for asthma and allergies, but there are some important aspects of the etiology of these illnesses that we are clear on:

- Substantial evidence suggests that the early periods of one's life (in *utero*, early infancy, pre-school years) are the primary developmental periods for asthma and related allergies.
- While heredity plays a very important role in the determination of asthma and allergies, so too does the environment (i.e., indoor and outdoor air, including both air pollution and especially environmental tobacco smoke).
- Some groups have been shown to be more vulnerable to asthma and related allergies, typically those with lower levels of socioeconomic status.

Gender also plays an important role. For example, while rates of asthma – particularly severe asthma – are higher in boy children than girl children, boys are much more likely to 'out grow' their asthma, and girls much more likely to develop asthma at the onset of puberty. Some researchers suspect a link with obesity.

Typically, asthma and related allergies can be managed with medication, good lifestyle choices, and the avoidance of triggers (e.g., environmental tobacco smoke, dust, moulds). However, according to the Asthma Society of Canada website (www.asthma.ca), about 500 Canadians die each year from asthma. Further, this burden of illness has substantial economic impacts on Canadian society; the Canadian Lung Association (www.lung.ca) estimates that respiratory diseases cost the Canadian economy approximately \$154 billion dollars annually.

A problem arises in the attempt to characterize the burden of illness for First Nations and Inuit communities in Canada, and this relates to a lack of data. Using what data we do have, as well as research literature characterizing similar populations, and extrapolating (cautiously) from what we know of the determinants of asthma and related allergies, we see a picture emerging that galvanizes the need for further investigation.

Worldwide, indigenous groups have been shown to be disproportionately represented at the severe end of the asthma spectrum (Chang 2007). In a study of emergency room visits for asthma and COPD, Sin et al (2002) found that Aboriginal Peoples are 2.1 times more likely to have an emergency room visit for asthma or COPD than their non-Aboriginal counterparts. Given what we know about the demographics and living situations of First Nations and Inuit communities in Canada, this may not be surprising. For example, we know that there are high rates of *tobacco use* among these communities; recent statistics indicate that the rate of smoking among First Nations people in Canada is 59%, approximately three times the rate of the general Canadian population. Smoking rates are even higher in the Inuvialuit population with approximately 70% of both men and women reporting that they smoke on a daily basis; 62% of Inuit women who were pregnant in 2001 reported smoking on a daily basis (Inuit Statistical Profile 2007). Smoking while pregnant and early childhood exposure to environmental tobacco smoke, have been shown to be key risk factors for asthma and related allergies.

Socioeconomic status has consistently been shown to be related to asthma and related allergies. We know that the median income for Aboriginal Peoples in Canada ranges widely but is typically less than \$40,000 per year (compared to \$55,000 for the rest of Canada in the same year). This would lead us to suspect that rates of asthma and related allergies may be higher in First Nations and Inuit communities. While affecting access to the determinants of health (e.g., adequate housing and nutrition, etc), lower income can also affect access to care. Despite access to universal health care in Canada, Ungar and Ariely (2005) have shown that 10% of Canadian children diagnosed with asthma do not receive medication (see also Subbarao et al., 2009). Further, an investigation by Anita Kozyrskyj (2001) of access to inhaled corticosteroids (a primary treatment for asthma) revealed that children covered by the First Nations benefits programs were only 82% as likely as children from a higher socioeconomic group to receive a prescription.

We have already seen that individuals in First Nations and Inuit communities are more likely to report living in *over crowded* or *substandard housing*. Both of these may be risk factors for asthma and related allergies in these populations (see for example Lawrence and Martin 2001). For example, respiratory infections are more likely to occur

when children live in substandard housing (dusty, damp, poorly heated) and more likely to pass these infections on when living in overcrowded conditions. It is well known that frequent respiratory infections in early life are a key risk factor for asthma and related allergies. However, it is important to note that there is conflicting evidence in the literature on the role of crowding and, to be blunt, dirt. That is, there is a substantial and growing literature around what is known as the 'hygiene hypothesis'. This stems from the fact that the highest rates of asthma in the world are in the western/developed nations where we have done away with most infectious diseases, rely heavily on antibiotics, and essentially sterilize the environments in which we raise our children. There is some evidence to suggest firstly that living with many siblings has a protective factor against asthma given that exposure to microbes helps young bodies build strong immune systems. The same argument is put forth around exposure to dirt, more generally. It must be noted that this is simply a hypothesis gaining increasing recognition but it has yet to be empirically substantiated.

In addition to environmental tobacco smoke, dust and moulds, *indoor air quality* in northern First Nations and Inuit communities can also be impacted by the use of wood burning stoves for heating, as well as activities related to curing and tanning. Guggisberg (2003) undertook a study of 400 households in Deline, NWT in order to examine the relationship between wood smoke and respiratory symptoms. He found that women who smoked, were exposed to wood smoke, and engaged in curing and tanning activities reported statistically significantly higher rates of respiratory symptoms than the comparison group. No significant relationships were found for males.

Life stress may also contribute to the development and expression of asthma and related allergies. This can be brought on by many factors – personal, emotional, economic. A growing body of evidence also links maternal stress with childhood asthma. We know that many First Nations and Inuit children are much more likely to live with a lone parent, grandparent or other relative (45%) than is the case in the general Canadian population (18%). Recent studies show that children whose caregivers report high levels of stress and have difficulties parenting are at greatest risk for asthma (Klinnert at al, 1994). Further, prolonged exposure to maternal distress during the first seven years of life has been shown to be associated with a 1.3-fold increased risk of asthma diagnosis (Kozyrskyj et al., 2008).

While the links with asthma and *obesity* are not yet well understood, there is an emerging literature which points to a clear association (see Ford 2005). In 2005, 35% of Canadian adults were overweight; 24% were obese. In 2002-03, 36% of First Nations people living on reserve were overweight; 36% were obese. These higher rates of obesity among First Nations translate down to children and youth, with rates two to three times higher than children and youth in the general Canadian population. There is also evidence of an association between GERD (gastroesophageal reflux disease) and asthma; asthma treatment has been shown to also treat symptoms of GERD (Harding, 2005). There is also evidence to support the link between asthma and allergies and sleep disorders (Léger, 2006)

Health Care Services

Inuit receive most of their health care services from provincial or territorial governments either directly or through transfer or land claim agreements. For specific initiatives, Health Canada's First Nations and Inuit Health Branch allocates funding to health programs and services for on-reserve and northern Inuit peoples, including: children and youth; chronic disease and injury prevention; communicable disease control; environmental public health and research; health governance and infrastructure support; mental health and addictions; primary care and supplementary health benefits.

Summary

While there is a clear need to collect more information, and have access to better quality, more timely and reliable data, this background discussion of asthma and related allergies and its associated risk factors indicates that First Nations and Inuit communities may indeed be at substantial risk. While some of the prevalence data may indicate otherwise, it is important to note that this may be the result of low levels of diagnosis and reporting, due to inadequate access to care, as opposed to actual levels of illness burden. In general, First Nations and Inuit communities are reporting inadequate access to care. For example, a Statistics Canada survey indicated that over one-third of the on-reserve First Nations population felt they had poorer access to health care services than the Canadian population. Barriers to access reported by the respondent group included: no doctor/nurse available in the area; inadequate care provided; and wait lists. With respect to asthma, Sin et al (2002) reports that Aboriginal Peoples with asthma were 55% less likely to see a specialist and 66% less likely to undergo spirometry than their non-Aboriginal counterparts.

The primary data collected from the focus groups and interviews, reported on in the next section, will help us to be able to start filling in some of these gaps.

Section 3: Results from Interview Data Analysis

Interview data were collected through telephone interviews (n=30) with relevant community health personnel (2 per each First Nations and Inuit region) in order to better understand the burden of illness and potential mechanisms for alleviating this burden in First Nation and Inuit communities. Tables are utilized to display frequencies in the data set; data is partial and reflects only the top three most mentioned responses.

Community Health

In order to get a sense of whether asthma and associated-allergies is considered a community priority, respondents were asked to describe what they perceived to be the most urgent health priorities in their communities. Both First Nation and Inuit communities identified suicide as the number one priority (50%), followed closely by chronic disease (40%), including prevalence of diabetes and cancer (see Table 1). Other health priorities (7%) included asthma and related respiratory illness.

Table 1

Top 3 Health Priorities (n=30)		
First Nations		
1. Suicide (60%)	1. Suicide (40%)	1. Suicide (50%)
2. Chronic disease (33%)	2. Chronic disease (47%)	2. Chronic disease (40%)
3. Other (7%)	3. Other (7%)	3. Other (7%)

In the context of asthma itself, respondents were undivided in what they perceived to contribute to high rates of asthma (as described in the environmental scan) in their community (see Table 2). 40% of respondents in both First Nation and Inuit communities identified air quality as the most central factor leading to high rates of asthma. Overall, 30% of respondents indicated the prevalence of both poor housing conditions and tobacco use. Taken together, these three factors were thought to account for the high rates of asthma in these communities.

Table 2

Factors contributing to high rates of asthma (n=30)		
First Nations (n=15)	Inuit (n=15)	Total (n=30)
Air quality (40%)	Air quality (40%)	Air quality (40%)
Tobacco use (40%)	Housing (40%)	Housing (30%)
Housing (20%)	Tobacco use (20%)	Tobacco use (30%)

Mechanisms linked to reducing rates of asthma correspond to factors contributing to high rates of asthma in Table 2. Addressing potential triggers of asthma in the home (57% in both communities) was seen as a strategy to mediate poor air quality shaped by dust, mold, pollen and smoking. Overcrowding and substandard housing conditions act to exacerbate situations in which families are dealing with asthma and associated-allergies. A second strategy to reduce rates of asthma was the implementation of public awareness and education, noted by 20% of respondents in First Nations and 33% in Inuit communities. Additionally, First Nations and Inuit respondents (13% in each community) acknowledged that improvement to air quality and the introduction of smoking cessation programs, respectively, would help to alleviate current rates of asthma.

Table 3

What would help to reduce rates of asthma? (n=30)		
First Nations (n=15)	Inuit (n=15)	Total (n=30)
Presence of potential triggers in the home (67%)	Presence of potential triggers in the home (47%)	Presence of potential triggers in the home (57%)
Public awareness/education (20%)	Public awareness/education (33%)	Public awareness/education (27%)
Air quality improvement (13%)	Smoking cessation (13%)	Air quality improvement (10%)

Although asthma is considered a community-based problem, some groups of individuals are impacted more than others (as shown in Table 4). Participants from both communities (80% overall) agreed that children were most affected by asthma, and that the elderly (7%) were also significantly affected; both groups were thought to be most vulnerable to asthma and associated-allergies.

Table 4

Who is most affected by asthma?		
	(n=30)	
First Nations	Inuit	Total
(n=15)	(n=15)	(n=30)
Children (73%)	Children (87%)	Children (80%)
Elderly (7%)	Elderly (7%)	Elderly (7%)
Other (13%)	Don't know (6%)	Other (7%)

When respondents were asked to describe existing barriers to accessing community resources, it was obvious there is a lack of culturally appropriate materials to increase public awareness and education (see Table 5). While 60% of First Nations and 40% of Inuit participants noted this barrier, another 20% (overall) felt that the overburdened health care system hindered access to asthma and allergy-related resources. 17% (overall) of respondents indicated that a transportation and jurisdiction policy also acted as an obstacle to pertinent resources.

Table 5

Exis	Existing barriers to accessing resources		
	(n=30)		
First Nations	Inuit	Total	
(n=15)	(n=15)	(n=30)	
Lack of culturally appropriate public awareness/education (60%)	Lack of culturally appropriate public awareness/education (40%)	Lack of culturally appropriate public awareness/education (50%)	
Jurisdictions/Transportation policy (20%)	Overburdened health care system (27%)	Overburdened health care system (20%)	
Overburdened health care system (13%)	Jurisdictions/Transportation policy (13%)	Jurisdictions/Transportation policy (17%)	

Community Resources

Although respondents seemed fairly unanimous in relationship to barriers that inhibit access to asthma and allergy-related resources, they also identified some resources that *were* available in their communities. In Table 6, we can see that in both communities respondents (57% overall) suggest that hospitals, public health nurses, and community health workers were considered the number one resource for individuals and families dealing with asthma and related-allergies. Programming for pregnant women and babies was considered by 17% (overall) of respondents, while free puffers and asthma medication was also noted (by 10% of all respondents).

Table 6

1 abic 0		
What resources exist in your community to address asthma?		
	(n=30)	
First Nations	Inuit	Total
(n=15)	(n=15)	(n=30)
Hospital/Public health nurse	Hospital/Public health nurse	Hospital/Public health nurse
/Community health worker	/Community health worker	/Community health worker
(60%)	(53%)	(57%)
Free puffers/Medication	Programming for pregnant	Programming for pregnant
(13%)	women/babies (20%)	women/babies (17%)
	, , ,	
Programming for pregnant	None (13%)	Free puffers/Medication
women/babies (13%)		(10%)

The types of resources illustrated by respondents to help children and families identify and manage their asthma largely reflect the documented barriers to accessing resources (see Table 5). In Table 7, 67% of respondents from First Nation communities and 60% from Inuit communities indicated that culturally appropriate asthma education and awareness would help children and families to identify and manage their asthma. Another 27% (overall) suggested that the services of primary health care providers and specialists would encourage the identification and management of asthma.

Table 7

What types of resources would help children/families identify and manage their asthma? (n=30)		
First Nations (n=15)	Inuit (n=15)	Total (n=30)
Culturally appropriate asthma education/awareness (67%)	Culturally appropriate asthma education/awareness (60%)	Culturally appropriate asthma education/awareness (63%)
Primary health care provider/Specialist (20%)	Primary health care provider/Specialist (33%)	Primary health care provider/Specialist (27%)
Don't know (13%)	Don't know (7%)	Don't know (10%)

When respondents were asked about whether families receive an asthma action plan when their children are diagnosed with asthma (Table 8), most (70% overall) indicated that they did not. Another 20% did not know whether this practice was offered; a very small proportion (7%) of respondents in First Nations communities was aware of the provision of asthma action plans upon diagnosis.

Table 8

Are you aware if families receive an asthma action plan when their children are diagnosed? $(n=30)$		
First Nations (n=15)	Inuit (n=15)	Total (n=30)
No (73%)	No (67%)	No (70%)
Don't know (20%)	Don't know (20%)	Don't know (20%)
Yes (7%)	Sometimes (13%)	Sometimes (7%)

In line with what respondents identified as barriers to accessing resources (see Table 5), in Table 9 respondents confirmed this assertion as 80% (in each community) believed that written materials for asthma are not tailored to address cultural sensitivities of communities. Another 10% (overall) felt that these materials were only somewhat tailored to cultural sensitivities, while 7% simply did not know.

Table 9

Are written materials for asthma tailored to address cultural sensitivities of your			
	community?		
	(n=30)		
First Nations	First Nations Inuit Total		
(n=15)	(n=15)	(n=30)	
No (80%)	No (80%)	No (80%)	
Somewhat (13%)	Yes (7%)	Somewhat (10%)	
Don't know (7%)	Somewhat (7%)	Don't know (7%)	

In addition to written materials, respondents were asked to describe their knowledge of clinical guidelines of community health professionals. Nearly half (47% overall) of all respondents acknowledged that health professionals in the community were *unaware* of clinical guidelines for asthma. 27% of the sample indicated that health professionals were only somewhat aware of clinical asthma guidelines, while another 16% simply did not know.

Table 10

Are health professionals in your community aware of clinical guidelines for asthma?		
Final Nations	(n=30)	Total
First Nations (n=15)	Inuit (n=15)	Total (n=30)
No (60%)	Somewhat (40%)	No (47%)
Don't know (20%)	No (33%)	Somewhat (27%)
Somewhat (13%)	Yes (13%)	Don't know (16%)

Burden of Illness

While respondents were acutely aware of the paucity of available and appropriate community resources, most did not know the scale of the burden of illness (see Table 11). 77% (between both communities) of respondents indicated that they were unaware of the number of hospitalizations in their community from asthma. Others (17% overall) did not know, while a smaller 6% (between both communities) indicated that they *were* aware of the number of community-hospitalizations.

Table 11

Are you aware of the number of hospitalizations from your community due to asthma?			
(n=30)			
First Nations	Inuit	Total	
(n=15)	(n=15)	(n=30)	
No (87%)	No (67%)	No (77%)	
Yes (7%)	Don't know (27%)	Don't know (17%)	
Don't know (6%)	Yes (6%)	Yes (6%)	

Further to Table 11, when asked whether or not a practice of tracking or collecting information of clients with asthma existed in their community, most respondents (70% overall) indicated that this process was not available (see Table 12). Another 27% of all respondents did not actually know if this practice existed, while only 3% (overall) knew of this practice.

Table 12

Is there any practice of tracking or collecting information of clients with asthma in your community?		
(n=15)	(n=15)	(n=30)
No (73%)	No (67%)	No (70%)
Don't know (27%)	Don't know (27%)	Don't know (27%)
	Yes (6%)	Yes (3%)

Community Expertise

In addition to resources (see Tables 6-10), respondents were asked to describe available community asthma expertise. More than half (60%) of all respondents indicated that there was actually no expertise available in their community, while 33% (overall) noted the availability of health care providers and ambulance personnel to address community asthma problems. A smaller 3% of respondents referred to the availability of community health/resource centres to address these types of problems in the community. Overall, only about a third of all respondents believed that there was actually expertise available to address asthma-associated problems in their community.

Table 13

Table 15			
Expertise available to address asthma problems in your community?			
(n=30)			
First Nations	Inuit	Total	
(n=15)	(n=15)	(n=30)	
None (73%)	Health care providers/Ambulance (47%)	None (60%)	
Health care providers/Ambulance (20%)	None (46%)	Health care provider/Ambulance (33%)	
Community health/Resource centres (7%)	Don't know (7%)	Community health/Resource centres (3%)	

An important predictor of asthma and allergy management and maintenance is the presence of family skills. Unfortunately, more than half (57% overall) of all respondents indicated that these types of skills were not present in family units. 16% (overall) of community members denoted the availability of education and information from health care providers, as well as the initial response (16%) to symptoms and asthma attacks (offered by ambulance personnel).

Table 14

Family skills to address asthma		
First Nations (n=15)	(n=30) Inuit (n=15)	Total (n=30)
None (47%)	None (67%)	None (57%)
Initial response to symptoms/attack (26%)	Education/Information from health care provider (20%)	Education/Information from health care provider (16%)
Education/Information from health care provider (13%)	Initial response to symptoms /attack (7%)	Initial response to symptoms /attack (16%)

Another important indicator of community resources, knowledge and education is the effective response of communities to other serious health problems. 50% of all respondents noted an effective community response to chronic disease (i.e. diabetes, cancer, etc.). In addition, another 40% (overall) felt that infectious disease was handled appropriately, while overall, a smaller 7% were unsure about the extent of how other health problems had been dealt with in their community.

Table 15

Other serious health problems that were addressed effectively (n=30)			
First Nations (n=15)	Inuit (n=15)	Total (n=30)	
Chronic disease (60%)	Infectious disease (47%)	Chronic disease (50%)	
Infectious disease (33%)	Chronic disease (40%)	Infectious disease (40%)	
Don't know (7%)	Housing conditions (7%)	Don't know (7%)	

In a final query, participants were asked to brainstorm about ideas that would help communities to address asthma in the future. Consistent with the sentiments of Tables 5 and 7, the majority of respondents (60%) indicated that culturally appropriate educational materials would be most effective in this capacity. In addition, regular check-ups and preventative programming was mentioned by 20% of all respondents to improve community responses to asthma. Another 10% indicated the need for relevant tobacco control by-laws to cope with circumstances related to asthma management and maintenance.

Table 16

Ideas to help address asthma in your community (n=30)		
First Nations (n=15)	Inuit (n=15)	Total (n=30)
Culturally appropriate educational materials (60%)	Culturally appropriate educational materials (60%)	Culturally appropriate educational materials (60%)
Regular check-ups /Preventative programming (20%)	Regular check-ups /Preventative programming (20%)	Regular check-ups /Preventative programming (20%)
Asthma workshops/Taskforce (7%)	Tobacco control by-laws (13%)	Tobacco control by-laws (10%)

Section 4: Results of Focus Group Analysis

Methods

Qualitative focus groups were conducted to elicit the perspectives of health care providers and community members on access to asthma and allergy care. Focus group data were coded according on the basis of key themes. Themes were generated both deductively from objectives outlined in the research proposal, and inductively as they emerged organically through focus groups. The key objective of the thematic analysis was to elaborate on and explore phenomena and processes at the individual level.

Key Themes

The results of focus group sessions are organized into key themes; lack of access to culturally appropriate asthma/allergy care and education, acknowledging the risk, and lack of community empowerment emerged from the thematic analysis. Community results are analyzed in order to describe Inuit and First Nation community's experiences of asthma and related allergies in the context of their surroundings and unique circumstances.

Theme 1- Lack of Access to Culturally Appropriate Asthma/Allergy Care and Education

Access to and uptake of asthma and allergy education and information varied among First Nations and Inuit communities. Levels of resources and knowledge of health care providers and community health workers was contextual and inconsistent. Families dealing with asthmatic children expressed frustration surrounding a lack of community resources and screening. In addition, respondents acknowledged the lack of culturally appropriate educational materials. In expressing concern regarding issues of language, many felt that interpretive services would allow them to become more informed about their child's asthma/allergy management.

In an effort to augment available community asthma/allergy resources, respondents illustrated the need for the implementation of community workshops and discussion forums as ways to engage community members in asthma/allergy education and consultation. Printed materials were also presented as ways "to help children to manage their asthma." Having access to reliable information was seen as key in providing parents and children with coping strategies at the onset of asthma attacks and allergy symptoms. The internet was seen as a potential source of uncertainty: "Internet is used a lot; you can pick up bad information, unreliable resources." Community workshops and education fairs would offer parents and children the opportunity to ask questions and to voice their concerns about dealing with asthma and allergies.

Respondents highlighted the need for hands-on education by having resources readily accessible in the community. Accessibility issues were paramount as parents felt that geography and weather often hindered their access to resources. They felt that parents and children:

Need to live with it; needs to be hands on education; if people don't really see and live with something they don't really understand it, so teaching needs to be hands on.

People want a reward or to be paid to participate in education; hard to get people together; distances make it hard too.

Concern surrounding the level of asthma/allergy knowledge of health care providers and asthma patients extended beyond asthma care, and intersected issues of geography.

I think the nurses need to be better educated as well. In a clinic in an isolated place they need to know what to do. My son had an allergic reaction and she (nurse) gave me an epipen and she didn't know how to use it.

This health centre is it- this is the only place that has the expertise; distant resources.

Respondents considered the relevance of the link between the dearth of culturally appropriate asthma/allergy education and the low levels of community diagnoses and underreporting. Low levels of diagnoses were common, and created uncertainty and worry in parents. Community perceptions of free medication were two-fold; free medication was considered positive for Inuit and First Nations people who were diagnosed with asthma and allergies, but was also considered negative when it was prescribed without a diagnosis.

Loose prescribing of asthma medications contributes to problems; everyone gets inhalers, even if no diagnosis of asthma.

This underreporting of allergies may be linked to the Doctor shortage; people are less likely to report as these are seen as less serious, not life threatening.

Respondents viewed this breakdown between education and diagnoses as evidence of inadequate access to health care services and medication.

People get the puffer prescribed and then discontinue use; don't use the medications properly, not being educated, don't understand the importance of taking the medication properly.

The potential consequence of improper medication-use was an additional worry of community members. Asthma-related policy development was identified as a means by which patients could access timely and appropriate care and instruction. Asthma and allergy screening was viewed as one method that might help to reduce symptoms before they become exacerbated. A framework of understanding was seen as necessary for proper asthma management; this framework was seen as being primary to encouraging asthma control and management in the community.

Education and protocols should be in place similar to protocols for TB so health professionals have a map to follow.

The only natural champions in the community are the community health reps, but this federal Health Canada program is ending and these roles are being lost.

School-related asthma and allergy policies were considered to be problematic from the stand-point of parents with asthmatic or allergic children. Inadequate asthma education for teachers has meant that children are responsible for their own asthma and allergy medication administration, regardless of their age and ability. Parents are encouraged to keep children at home if they are experiencing symptoms of asthma and or allergies; parents also felt that they were responsible for advocating for their children. The lack of school-related asthma support also resulted in parents feeling powerless in the management of their child's asthma and allergies. This lack of support translates into higher school absences, which has the potential to generate long-term implications for learning.

The policy requires that children sign papers, get medical forms filled out by doctor to allow medications to be kept at the school, but the staff cannot administer the medications.

There is a policy in the schools concerning asthma, but it doesn't encourage people to send their children to school when they are having active asthma symptoms; one woman described how she just keeps her children home when they are sick with asthma symptoms. The schools require the children to administer the medications themselves and her children are too young to do so.

Teachers aren't trained at all in asthma or allergy treatment.

The notion that children were unable to participate in physical activities was a source of regret for some parents. This regret appeared to have stemmed from the positive implications of physical activity that their children might have been missing. On the contrary, some respondents sensed that their children sometimes took advantage of their asthma and, as such, opted out of or simply refused physical activity. Using their diagnosis as an excuse not to participate in exercise was another example of how proper, accurate asthma education could help to enlighten parents and children of the health benefits of physical activity for asthmatics.

One effect in eldest child; used asthma diagnosis not to be physically active; can be used as an excuse.

Theme 2- Acknowledging the Risk

Respondents viewed asthma/allergy risk as inherent within their community landscape; the most common triggers were: living/housing conditions; smoking; and chronic disease. Respondents felt that children and elders were most at risk for respiratory illness in their community. In addition, babies were also seen as facing risk due to the smoking habits of their mothers and other family members.

Housing conditions were consistently viewed as one of the most common variables contributing to rates of asthma and allergies in these communities. Dust, pollen, pets, mould and smoking were thought to both exacerbate existing symptoms, and to increase asthma incidences in both adults and children. Overcrowding was common among these communities, as evident in the literature above describing cultural cohabitation practices. Respiratory illness can become transmitted via crowded and unsanitary living conditions, which can lead to allergic sensitivities and increased rates of asthma and related respiratory conditions. Living conditions were viewed as a source of frustration, and respondents felt a sense of powerlessness in an effort to improve them.

One story told of how a brand new house developed mould because the insulation was allowed to get wet during the construction phase.

One respondent expressed that they: "still have asbestos insulation in many homes on the community."

Improper inspections of houses, built on swamps, landfill sites, water run through where people live.

Lots of mould in homes, in bathrooms, kitchens, as a result of leaks, improper insulation, construction defects.

Respondents viewed home carpeting as a major contributor to allergy exacerbation. Cleanliness was linked to carpeting as it was difficult to rid the carpeting of smoke, dirt and dust. Dust from gravel roads contributed to this problem during construction when

ground cover becomes destroyed by construction vehicles. Respondents indicated that exposure to strong scents within and outside of the home, such as perfumes, negatively impacted allergic sensitization.

As indicated, tobacco use in Inuit and First Nations communities is substantially higher than that of the general population; respondents were to some extent aware of the impact of tobacco use (including second-hand smoke) and poor air quality on rates of asthma and allergies.

I smoked 16 years ago, and after I quit smoking it was a big difference- you could never believe. I don't think people realize if you quit smoking or if you're smoking how much sicker you can get.

Respondents differentiated between the consequences of indoor and outdoor smoking, noting that outdoor smoking lead to improvements in the respiratory health of children in the home.

Early onset of smoking, and in non ventilated places, smoking near kids. Child with no puffers for 2 weeks in a house with smokers and wood fire had to be taken away from the home to have a chance to get better.

My two boys are asthmatic and I am as well. It was hard to figure out; one was diagnosed at 6 months and one at 3 months. The oldest outgrew it but the youngest one still – he's 15 now and still has it. I am a smoker. Yes, that is a problem but I quit when I was pregnant with my second child and his asthma is worse.

Although there were smoking cessation programs in select communities, respondents were undivided in their support for improving and increasing community smoking cessation programming, especially targeted at pregnant women. They felt strongly that there was hope for improvement in respiratory health if parents and youth were informed about the direct implications of smoking for lung health, for example "talk with people that have quit, or visit with people with cancer patients." In addition, implementing a blue-light initiative for smoke-free homes as well as limiting indoor smoking in office areas was suggested as ways to encourage smoking cessation.

Indoor wood stoves and outdoor environmental pollution were also seen as indicators of poor air quality, exacerbating existing asthma and allergy symptoms. The use of air exchangers was minimal; however, there was hope that air quality testing might facilitate a reduction in rates of asthma and allergies.

Woodstoves are used in most homes, affecting breathing.

The fallout from the mill smoke has set off home fire alarms or carbon monoxide alarms, even up to 20 minutes away.

A small number, about 2% of homes use air exchangers, but they break down and people don't repair them, don't use them and leads to mould build up.

The prevalence of chronic disease in Inuit and First Nations communities was manifested in perceived rates of cancer and diabetes, as "cancer and diabetes seem to be a big thing here." Some respondents felt that these issues had been addressed appropriately: "it was addressed effectively through Diabetes Prevention workers." For others, the risk of chronic disease did not seem to be mediated by acceptable and timely health care and/or health promotion programming. Respondents expressed similar frustration for the paucity of education and programming surrounding community chronic disease.

All other major health problems have the same history of not being dealt with appropriately. For example, diabetes and high blood pressure show the same kind of issues/problems.

Theme 3- Lack of Community Empowerment

Lack of community empowerment was broadly illustrated by respondents. A variety of obstacles prevent communities from feeling as though they are able to take control and manage their asthma, including funding, transportation, and subsequently, isolation. Respondents felt that their asthma was controlled by variables that operated as barriers in the context of asthma and allergy diagnosis, management and education and information.

Governmental funding was a source of tension as many respondents highlighted the funding disparities existing between their communities versus the general population.

There are vast disparities in service being allotted to First Nation population versus the CCAC home care for the general population. First Nations are a population that has a much higher incidence of asthma and mortality rate than the average population. This could be addressed at a higher level to attain relative parity of funds and service. Should have an equal service to all First Nations people, not just the people who are status on reserve. A united health care program for all status clients would be the ideal (First Nation Home Care would deal with all First Nation people, on and off reserve).

Need is greater despite the population, yet same funds or less per capita for home care program.

As outlined, the burden of illness in Inuit and First Nations communities is greater than that of the general population. Rates of asthma and allergy were attached to the lack of access to health care in general:

Lack of money for nurses in the community.

Wait in emergency unit is long. If it is not life threatening, the reception is very surly, not welcoming.

Respondents connected the lack of funding for asthma and allergy care to a sense of powerlessness. As treatment continues to be unavailable, the situation is perceived to be isolating and urgent. Uncertainty prevails as community members struggle to manage asthma and allergy symptoms, in addition to their living environment.

Things are getting worse- 20 years ago medical concerns were not an issue.

The nature of these remote communities makes transportation problematic and costly, and access to timely health care in emergency situations can become challenging. The distribution of the population is affected by weather for most of the year, which makes accessing primary and secondary health care difficult.

3 out of 10 communities are so far away, it can take around 4 hours to get to the hospital after someone attends to your child.

You're in isolation and in order to get to a bigger centre you have to travel to a bigger centre, you have to go through your doctor to get tested for allergies. How many people with asthma ever saw a respiratory therapist in how to use a puffer correctly? You have allergies/asthma so you live with it.

Have to medivac people out; helicopter or land ambulance. Two staff members from the community have to meet the government helicopter.

Some communities are 75km just to the highway (remote for the land ambulance). If you have the ability to come into the clinic you can get the education, but you need the money to come in, vehicle, time.

You might have to go to St. John's or Goose Bay to get tested and if you don't get past your Doctor then you don't go. We can't afford to go to St. John's on our own.

Uncertainty was prevalent as funding for asthma and allergy care continues to impact the everyday lives of participants and their families. Respondents felt that the circumstances surrounding asthma and allergies were much more than what they could cope with without assistance and screening from accredited health care providers. In the context of Inuit and First Nations communities, a framework for asthma and allergy care would be advantageous in that it would provide access to health care providers for increased diagnoses, information and education, and appropriate patient follow-up and monitoring.

Section 5: Conclusions and Recommendations

This exploratory study of the perceptions and experiences of asthma and related allergies in First Nations and Inuit communities in Canada has only begun to scratch at the surface of the issues in prevalence, impacts, and the need for management strategies tailored to these communities. However, the data reported herein provides a strong foundation for building a response, and for providing a way forward.

These data have shown us firstly that there are major gaps in family and community resources for coping with and managing asthma and related allergies. Further, there appears a lack of professional health care resources, opportunities and training as well; not surprising due to a range of access issues, including geographic access. Further, broad issues such as adequate housing and community support through the school system must be addressed in the face of this growing public health issue.

It is also important to point out the limitations of the study. Given its exploratory nature, perhaps there might have been additional questions we should have asked, or questions that could have/should have been asked in a different way. For example, the ITK felt strongly that asthma and related allergies would have emerged as a higher priority if the question had been targeted to children and youth. Further, the sample size employed in the study is small; while it is important to remember that the study was not meant to be representative, a larger sample size would have allowed for more sophisticated statistical analysis. Finally, it would have been useful to tape record the surveys as they were administered as well as the focus groups. This would have allowed a more comprehensive analysis of the data as collected. Regardless of these limitations, however, the data provide a useful snapshot of the issues related to asthma and related allergies in these communities and further allow us to make some informed recommendations.

The National Lung Health Framework recently released a draft action plan for consultation. This draft action plan acknowledges alarming rates of respiratory disease among First Nations, Inuit and Metis people in Canada and indicates that these communities are disproportionately affected by inequities in health outcomes as a result of poverty and social inequality. The action plan prescribes a number of key strategies and activities which should be implemented over the next five years. When implemented, all of these will result in improved respiratory health for all Canadians including First Nations and Inuit but several key activities relate specifically to these communities. Examples of these activities include: develop and implement awareness initiatives targeting chronic disease, with an emphasis on at-risk populations; promote collaborative community-based interventions that address local environmental exposure issues where everyone in Canada lives, learns, works and plays; expand evidence to identify and confirm the factors that place certain populations at increased risk and develop and implement early detection programs that reflect population-based risk factors that focus in these at-risk populations; identify at-risk populations and the barriers that are preventing their access to respiratory health and create/develop programs to address

these barriers to access; and identify the impact that investments in community programs have on health outcomes with a focus on at-risk populations.

The findings from both phases of data collection and analyses support five key recommendations.

- 1. The development of culturally appropriate education and community asthma/allergy awareness is integral for improving access to asthma and associated-allergy related health services. Translating these types of knowledge to Inuit and First Nations communities would first require funding to initiate partnerships between policy makers and community working groups to ensure that materials are sensitive to a range of community needs in order to mobilize communities to better understand and manage their asthma, related-allergies and lung health. Translation of educational materials into appropriate languages would help ensure that resources are both culturally-sensitive and comprehensive. Access to asthma and allergy care and education would encourage community empowerment, and provide a knowledge-base for community members to be able to cope with and manage their symptoms for an improved quality of life. Targeting specific learning strategies for community health workers working in First Nation and Inuit communities would be valuable.
- 2. There is an acute need for a federal framework of asthma and allergy care for health care providers. In order to improve the level of diagnosis, care and management, physicians, nurses and community health workers should be mandated to follow government asthma/allergy guidelines specific to Inuit and First Nation communities. In consultation with the National Lung Health Framework, such a structure would be funded, guided and developed in partnership with policy makers and community working groups to make certain that the needs of the communities are the basis upon which it is designed and implemented. Strengthening partnerships at provincial and territorial (First Nation and Inuit organizations) levels between asthma and lung association organizations is an important component of ensuring understanding at the regional level. The quantitative data supports a lack of resources, opportunities and training available to health care professionals in the communities. It would be important to facilitate a discussion of whom the relevant health care professionals are in the community and in what ways they can be supported, in addition to assessing the availability of materials and educational strategies available to community health professionals.
- 3. There is a consensus around the need for the development of a new health care delivery model for First Nation and Inuit peoples with chronic conditions. This recommendation is supported by the identification of asthma-related policy development as a means by which patients and their families could access timely and appropriate care and instructions.

- 4. We must begin to address major gaps in family and community resources for managing and coping with asthma and related allergies. Integral to this finding is the implementation of programs for chronic disease prevention, health promotion, tobacco control and prevention, nutrition and obesity, and programs for asthma, lung health and related-allergies. Asthma is unique as a chronic disease and managing and coping requires appropriate resource allocation at the family and community level. High smoking rates reported in the environmental scan support a need for a tobacco reduction strategy in First Nation and Inuit communities. For example, the interaction of various environmental risk factors makes children especially vulnerable to asthma and related- allergies as they experience high rates of acute respiratory infections, and exposure to environmental tobacco smoke and dust and molds. Housing issues are also linked to the National Lung Health Framework as the relationship between indoor air quality and respiratory health requires further exploration in the context of crowded dwellings, housing repairs, and access to garbage collection, septic and sewage services. In addition, as noted in the environmental scan, social groups that have lower levels of socioeconomic status tend to be more vulnerable to asthma and relatedallergies, which makes these communities particularly at risk and, therefore, a top priority for asthma and related-allergy programming.
- 5. There must be more attention paid to the link between obesity and asthma and related-allergies. It would be valuable to provide information and educational programs that address asthma and obesity issues in the context of nutrition and physical activity. School-based programs for children with asthma and related allergies are one way to integrate education and awareness through resources and training at the local level. Given the findings of the current study, a key strategy would be to target school-aged children and teachers to increase their knowledge about lung health, asthma and allergies in health and physical education classes. A program of this nature would increase teacher understanding of lung health and offset the misconceptions of exercise as a danger to asthma and allergies, as well as place some of the responsibility of physical activity onto the school boards.

Cutting across these recommendations is a clear need for more research, better data, increased monitoring and surveillance, and enhanced access to care. Implicit across these recommendations is the recognition of the importance of the social determinants of health to the quality of life and well being of Canada's First Nations and Inuit peoples. There must be explicit recognition of these in any long-term research agenda.

References

- Assembly of First Nations (2007). First Nations regional longitudinal survey (RHS) 2002/03: results for adults, youth and children living in First Nations communities. Retrieved March 15, 2008, from http://www.rhsers.ca.english/phase1.asp
- Bai, T., Kenney, A. & Van Gorder, B. (2001). Asthma. In V. Bryanton, Y. Chen, H. Johansen, K. Leeb, L. McRae, P. Michaelson, C. Nair, & P. Stewart. *Respiratory Diseases in Canada* (pp. 33-44). Ottawa, ON: Health Canada.
- Berghout, J., Miller, J.D. Mazerolle, R., O'Neill, L. Wakelin, C., MacKinnon, B., Maybee, K., Augustine, D., Levi, C.A., Levi, C., Levi, T., & Milliea, B. (2005) Indoor environmental quality in homes of asthmatic children on the Elsipogtog reserve (NB), Canada. *International Journal of Circumpolar Health*, 64(1), 77-85.
- Boulet, L.P., Becker, A., Berube, D., Beveridge, R., & Ernst, P. (1999). Canadian asthma consensus report, 1999. *CMAJ*, 161(Suppl. 11), 1-62.
- Brazil, K., & Krueger, P. (2002). Patterns of family adaptation to childhood asthma. *Journal of Pedatric Nursing*, 17, 167-173.
- Cakmak, S., Dales, R.E., Judek, S., & Coates, F. (2005). Does socio-demographic status influence the effects of pollens and molds on hospitalization for asthma? Results from a time-series study in 10 Canadian cities. *Annals of Epidemiology*, 15(3), 214-218.
- Canadian Allergy, Asthma and Immunology Foundation. (2006) *Allergies and asthma:* our research helping you and your loved ones. [Brochure]. Ottawa, ON.
- Chang, A.B., Taylor, B., Masters, I.B., Laifoo, Y., & Brown, A.D.H. (2007). Indigenous healthcare worker involvement for Indigenous adults and children with asthma. *Cochrane Database of Systematic Reviews*, 4, Article No.: CD006344. DOI: 10.1002/14651858. CD006344.pub2.
- Chatzi, L., Torrent, M., Romieu, I., Garcia-Estaban, R., Ferrer, C., Vioque, J., Kogevinas, A., & Sunyer, J. (2008). Mediterranian diet in pregnancy is protective for wheeze and atopy in childhood. *Thorax*, 20, 1-8.
- Dales, R.E., Choi, B., Chen, Y., & Tang, M. (2002). Influence of family income on hospital visits for asthma among Canadian school children. *Thorax*, *57*, 513-517.
- Earl, S.F. (2005). The epidemiology of obesity and asthma. *Journal of Allergy and Clinical Immunology*, 115, 897-909.

- First Nations and Inuit Health Committee, Canadian Paediatric Society. (2006). Use and misuse of tobacco among Aboriginal Peoples update 2006. *Paediatrics & Child Health*, 11(10), 681-685.
- Fitzsimon, N., Fallon, U., O'Mahony, D., Loftus, B.G., Bury, G., Murphy, A.W., & Kelleher, C.C. Mothers' dietary patterns during pregnancy and risk of asthma symptoms in children at 3 years. *Irish Medical Journal*, 100(Suppl. 8), 27-32.
- Fletcher, C.M., & Pride, N.B. (1984). Definitions of emphysema, chronic bronchitis, asthma, and airflow obstruction: 25 years from the Ciba symposium. *Thorax*, 39(2), 81-85.
- Gehring, U., Pattenden, S., Slachtova, H., Antova, T., Braun-Fahrländer, C., Fabianova, E., Fletcher, T., Galassi, C., Hoek, G., Kuzmin, S.V., Luttman-Gibson, H., Moshammer, H., Rudnai, P., Zlotkowska, R., & Heinrich, J. (2006). Parental education and children's respiratory and allergic symptoms in the Pollution and the Young (PATY) study. *European Respiratory Journal*, 27(1), 95-107.
- Guggisberg, M., Hessel, P.A., Michaelchuk, D., & Ahmed, I. (2003). Respiratory symptoms and exposure to wood smoke in an isolated northern community. *Canadian Journal of Public Health*, *94*(5), 372-376.
- Guggisberg, M., Hessel, P.A., Michaelchuk, D., & Atiemo, M. (2003). Particulate matter and gaseous contaminents in indoor environments in an isolated northern community. *International Journal of Circumpolar Health*, 62(2), 129-129.
- Harding, S.M. (2005). Gastroesophageal reflux: a potential asthma trigger. *Immunology* and Allergy Clinics of North America, 25(1), 131-148.
- Health Canada (2005). First Nations, Inuit and Aboriginal health: statistical profile on the health of First Nations in Canada. Retrieved July 29, 2008, from http://www.hc-sc.gc.ca/fniah-spnia/pubs/aborig-autoch/stats_profil-eng.php
- Health Canada. (2006). *Healthy Canadians: A federal report on comparable health indicators*. (HC Pub.: 1192). Ottawa, ON: Health Canada Publications.
- Health Canada. (2008). First Nations, Inuit and Aboriginal health: funded health programs and services. Retrieved November 24, 2008 from http://www.hc-sc.gc.ca/fniah-spnia/finance/agree-accord/prog/index-eng.php
- Hemmelgarn, B., & Ernst, P. (1997). Airway function among Inuit primary school children in far northern Quebec. *American Journal of Respiratory and Critical Care Medicine*, *156*(6), 1870-1875.

- Ho, L.S., Gittelsohn, J., Harris, S.B., & Ford, E. (2006). Development of an integrated diabetes prevention program with First Nations in Canada. *Health Promotion International*, 21(2), 88-97.
- International consensus report on diagnosis and treatment of asthma, 1992. National Heart, Lung and Blood Institute, National Institutes of Health. Bethesda, Maryland. *European Respiratory Journal*, *5*(5), 601-641.
- Kozyrskyj, A.L., Mustard, C.A., & Simons, F.E. (2001). Socioeconomic status, drug insurance benefits, and new prescriptions for inhaled corticosteroids in school children with asthma. *Archives of Pediatrics & Adolescent Medicine*, *155*, 1219-1224.
- Kozyrskyj, A.L., Mustard, C.A., Cheang, M.S., & Simons, F.E.R. (2001). Income-based drug benefit policy: impact on receipt of inhaled corticosteroid prescriptions by Manitoba children with asthma. *Canadian Medical Association Journal*, 165(7), 897-902.
- Kuhnlein, H.V., Receveur, O., Soueida, R., & Egeland, G.M. (2004). Arctic Indigenous peoples experience the nutrition transition with changing dietary patterns and obesity. *Journal of Nutrition*, 134, 1447-1453.
- Lambden, J., Receveur, O., Marshall, J., & Kuhnlein, H.V. (2006). Traditional and market food access in Arctic Canada is affected by economic factors. International *Journal of Circumpolar Health*, 65(4), 331-340.
- Lawrence, R., & Martin, D. (2001) Moulds, moisture and microbial contamination of First Nations housing in British Columbia, Canada. *International Journal of Circumpolar Health*, 60(2), 150-156.
- Léger, D., Annesi-Maesan, I., Carat, F., Rugina, M., Chanal, I., Pribil, C., Hasnaoui, A.E., & Bousquet, J. (2006). Allergic rhinitis and its consequences on quality of sleep: an unexplored area. *Archives of Internal Medicine*, *166*(16), 1744-1748.
- Leung, D.Y.M., Boguniewicz, M., Howell, M.D., Nomura, I., & Hamid, Q.A. (2004). New insights into atopic dermatitis. Journal of Clinical Investigation, 113(5), 651-657.
- Norris, M.J. (2007). Aboriginal languages in Canada: emerging trends and perspectives on second language acquisition (Catalogue No. 11-008). *Canadian Social Trends*, 83. Ottawa, ON: Statistics Canada.
- Pauktuutit Inuit Women of Canada. (2006). The Inuit way: a guide to Inuit culture. Retrieved March 15, 2008, from http://www.pauktuutit.ca/pdf/publications/pauktuutit/InuitWay_e.pdf.

- Royal Commission on Aboriginal Peoples (RCAP). (1996). Report of the Royal Commission on Aboriginal Peoples. Ottawa, ON: Indian and Northern Affairs Canada.
- Saksvig, B.I., Gittelsohn, J., Harris, S.B., Hanley, A.J.G., Valente, T.W., & Zinman, B. (2005). A pilot school-based healthy eating and physical activity intervention improves diet, food knowledge, and self-efficacy for native Canadian children. *Journal of Nutrition*, 135, 2392-2398.
- Seguin, L., Xu, Q., Gauvin, L., Zunzunegui, M.V., Potvin, L., & Frohlich, K.L. (2005). Understanding the dimensions of socioeconomic status that influence toddlers' health: unique impact of lack of money for basic needs in Quebec's birth cohort. *Journal of Epidemiology and Community Health*, *59*(1), 42-48.
- Senthilselvan, A., Lawson, J., Rennie, D.C., Dosman, J.A. (2003). Stabilization of an increasing trend in physician-diagnosed asthma prevalence in Saskatchewan, 1991 to 1998. *Chest*, 124, 438-448.
- Sharma, S., Cao, X., Gittelsohn, J., Ho, L.S., Ford, E., Rosecrans, A., Harris, S., Hanley, A.J., Zinman, B. (2007). Dietary intake and development of a quantitative food-frequency questionnaire for a lifestyle intervention to reduce the risk of chronic diseases in Canadian First Nations in north-western Ontario. *Public Health Nutrition*, 11(8), 831-840.
- Shephard, R.J. (2007). Fitness of Canadian children: range from traditional Inuit to sedentary city dwellers, and assessment of secular changes. *Medicine and Sport Science*, 50, 91-103.
- Signer, A., & Costa, R. (2005). Aboriginal conditions in census metropolitan areas, 1981-2001, (Catalogue No. 89-613-MIE). Ottawa, ON: Social and Aboriginal Statistics Division, Statistics Canada.
- Sin, D.D., Spier, S., Svenson, L.W., Schopflocher, D.P., Senthilselvan, A., Cowie, R.L., Man, S.F.P. (2004). The relationship between birth weight and childhood asthma: a population-based cohort study. *Archives of Pediatrics and Adolescent Medicine*, 158, 60-64.
- Sin, D.D., Wells, H., Svenson, L.W., & Man, S.F.P. (2002). Asthma and COPD among Aboriginals in Alberta, Canada. *Chest*, *121*(6), 1841-1846.
- Skinner, K., Hanning, R.M., & Tsuji, L.J. (2006). Barriers and supports for healthy eating and physical activity for First Nation youths in northern Canada. *International Journal of Circumpolar Health*, 65(2), 148-161.
- Statistics Canada (2003). Aboriginal Peoples survey 2001 initial findings: well-being of the non-reserve Aboriginal population, (Catalogue No. 89-589-XIE). Ottawa, ON.

- Statistics Canada (2004). A portrait of Aboriginal children living in non-reserve areas: results from the 2001 Aboriginal peoples survey, (Catalogue No. 89-597-XIE). Ottawa, ON.
- Statistics Canada (2005, June 28). Canada's Aboriginal population in 2017. *The Daily*. Retrieved on July 29, 2008, from http://www.statcan.ca/Daily/English/050628/d050628.htm
- Story, R.E. (2007). Asthma and obesity in children. *Current Opinion in Pediatrics*, 19(6), 680-684.
- Subbarao, P., Becker, A., Brook, J.R., Daley, D., Mandhane, P.J., Miller, G., Turvey, S.E., & Sears, M.R. (in press 2009). Epidemiology of asthma: risk factors for development.
- Tait, H., Nepton-Riverin, M., & Clark, C. (2007). Inuit in Canada: a statistical profile. Retrieved March 15, 2008, from http://www.itk.ca/Inuit-Statistical-Profile
- Ungar, W.J., & Ariely, R. (2005). Health insurance, access to prescription medicines and health outcomes in children. *Expert Review of Pharmoeconomics & Outcomes Research*, 5(2), 215-225.
- Wilkins, R., Uppal, S., Finès, P., Senécal, S., Guimond, É., & Dion, R. (2008, January 23). Life expectancy in the Inuit-inhabited areas of Canada, 1989 to 2003. *The Daily*. Retrieved on November 22, 2008, from: http://www.statcan.ca/english/freepub/82-003-XIE/2008001/article/10463/finding-en.htm.
- World Allergy Organization. (2006). Allergic Diseases Resource Centre. Available at http://www.worldallergy.org/adrc.
- Young, T.K., & Katzmarzyk, P.T. (2007). Physical activity of Aboriginal Peoples in Canada. *Applied Physiology, Nutrition, and Metabolism, 32*(Suppl. 2E), 148-160.
- Young, T.K. (2003). Review of research on Aboriginal populations in Canada: relevance to their health needs. *British Medical Journal*, 327, 419-422.
- Zindah, M., Belbeisi, A., Walke, H., & Mokdad, A.H. (2008). Obesity and diabetes in Jordan: findings from the behavioral risk factor surveillance system, 2004. *Preventing Chronic Disease*, 5(1), 1-8.

APPENDICES

Appendix 1: Survey Questions

PRIORITIES

- 1. Tell me about the health priorities in your community. What would you say are the top three?
- 2. What factors do you think contribute to that high rate of asthma in your community? Allergies?
- 3. What would help to reduce the asthma rate in your community? Allergies? Homes? (probe—air quality, mould, smoking) Schools?

GAPS

- 4. Who is most affected by asthma in your community? Allergies?
- 5. What are the existing barriers to accessing resources?

To Diagnosis?

To Treatment?

To Intervention?

- 6. What resources exist in your community to address asthma? Allergies?
- 7. What type of resources would help families to identify asthma/allergies in their children?
- 8. Are you aware if families receive Asthma Action Plans when their children are diagnosed?
- 9. Are written materials for asthma tailored to address cultural sensitivities of your community? Allergies?
- 10. Are health professionals in your community aware of clinical guidelines for asthma?
- 11. Are you aware of the # of hospitalizations from your community due to asthma? Allergies?
- 12. Is there any practice of tracking or collecting information of clients with asthma in your community? Allergies?

If yes, what do you collect? (Number of clients, age, sex, years with asthma, etc)

ASSETS

- 13. What expertise is available to address asthma problems in your community? Allergies?
- 14. What skills do families have to address asthma? Allergies?
- 15. Was there another serious health problem that was addressed effectively in your community?
- 16. What innovative ideas do you have that would help to address asthma in your community? Allergies?

Appendix 2: Focus Group Questions

COMMUNITY PRIORITIES

- 1. Tell me about the health priorities in your community. What would you say are the top three?
- 2. What factors do you think contribute to that high rate of asthma in your community? Allergies? Why do you think that is?
- 3. What would help to reduce the asthma rate in your community? Allergies?
- 4. Do you have a child /children with asthma? Allergies? What did you understand about asthma when your child was first diagnosed? Allergies?

GAPS

- 5. Who is most affected by asthma in your community? Allergies?
- 6. What are the existing barriers to accessing resources?

To Diagnosis?

To Treatment?

To Intervention?

- 7. What resources exist in your community to address asthma? Allergies?
- 8. Has asthma caused any limitations for your child? Allergies? How has this affected your family?
- 9. What type of resources would help you to identify asthma/allergies in your child? What is needed to help your child/ or other children in the community to manage their asthma? Allergies?

ASSETS

- 10. What expertise is available to address asthma problems in your community? Allergies?
- 11. What skills do families have to address asthma? Allergies?
- 12. Was there another serious health problem that was addressed effectively in your community? (Facilitator can provide context/probe for former examples of advocating for services/gaining knowledge, etc.)How?

13. What innovative ideas do you have that would help to address asthma in your community? Allergies?

SPECIFIC QUESTIONS

- 14. For those of you that have a child with asthma, do you remember getting the diagnosis? How did knowing you have a child with asthma make you feel? Allergies?
- 15. What's it like having a child with asthma? What's it like having a child with allergies?
- 16. What interactions or experiences have you had with school(s); tell me about that?
- 17. Does your child's teacher play any role in his/her asthma? Allergies?
- 18. Have you ever had a negative interaction with the school around your child's asthma? Allergies?
- 19. Do you think your child's asthma affects her/his daily life at school? Allergies?
- 20. Are you concerned about your child's asthma while he/she is at school? Allergies?

EXPERIENCES

- 21. Do you think you treat your child differently because of his/her asthma? Allergies?
- 22. Has your child felt excluded because of asthma? Allergies?
- 23. Has your child had an adverse reaction since his/her diagnosis?
- 24. Does your child always carry an inhaler? EpiPen?
- 25. Has your child ever forgotten to take his/her inhaler? EpiPen?
- 26. Has your child ever had to use his/her inhaler while at school? EpiPen?